Author’s response to reviews

Title: The effect of disclosure on adherence to antiretroviral therapy among adults living with HIV in Ethiopia: a systematic review and meta-analysis

Authors:
Getenet Dessie (ayalew.d16@gmail.com)
Fasil Wagnew (fasilowa2@yahoo.com)
Henok Mulugeta (mulugetahenok68@gmail.com)
Dessalegn Amare (desalegnezelellw@gmail.com)
Dube Jara (jaradube@gmail.com)
Cheru Tesema Leshargie (chertesema@gmail.com)
Ayenew Negesse (ayenewnegesse@gmail.com)
Swati Rayasam (rayasam.swati@gmail.com)
Sahai Burrowes (Sahai.Burrowes@tu.edu)

Version: 2 Date: 28 May 2019

Author’s response to reviews:
Dear Acadmic editor ,the individual contributions of Dessalegn Amare is included. The abbreviation "DA" stands for Dessalegn Amare.

Thank you!!!